Author’s response to reviews

Title: Surgical management of pyelo-ureteral junction syndrome in a resource-limited setting: Case of Zinder National Hospital, Niger

Authors:
Harissou Adamou (harissou1976@yahoo.fr)
Maazou Halidou (halidou3@yahoo.fr)
Ibrahim Amadou Magagi (imagagi@gmail.com)
Hassane Diongolé (diongolen@yahoo.fr)
Mahamadou Doutchi (m.doutchi@yahoo.fr)
Oumarou Habou (bhomar70@yahoo.fr)
Kabirou Ganiou (adениянakeji@yahoo.fr)
Amadou Soumana (drSoumanaamadou@gmail.com)

Version: 2 Date: 12 Dec 2018

Author’s response to reviews:

Answers to Reviewers

Dear editors, Thanks for the corrections made for the improvement of our manuscript. All corrections are incorporated in this revised version.

1. The names of the authors were reversed in the following order: Given names and surnames, instead of surnames and first names
2. Abstract:
   - The abstracts have been standardized.
3. Methods:
   - Changes were made to the methodology to talk about the study framework and to describe the catchment area of Zinder National Hospital.
   - The renal lodge is the renal compartment, not the renal pelvis.
   - The statement regarding consent/ethical approval was included.
4. Results:
   - The table was changed and a column was inserted for creatinine levels at 6 months of follow-
up.

- For the only case of anastomotic fistula, the retro-peritoneal drain was kept in place for 25 days (Table)
- A column was inserted for creatinine levels at 6 months of follow-up
- Cases of surgical site infection resolve with oral antibiotics only.

5. Discussion:
- The limits of the study were added in the first paragraph of the discussion.